Author's response to reviews

Title: Type I Gaucher disease with exophthalmos and pulmonary arteriovenous malformation

Authors:

Chun-An Chen (chenca@ntu.edu.tw)
Nelson LS Tang (nelsontang@cuhk.edu.hk)
Yin-Hsiu Chien (chienyh@ha.mc.ntu.edu.tw)
Wei-Min Zhang (nelsontang@cuhk.edu.hk)
Jou-Kou Wang (jkw@ha.mc.ntu.edu.tw)
Wuh-Liang Hwu (hwu@ha.mc.ntu.edu.tw)

Version: 6 Date: 27 May 2005

Author's response to reviews:

1. References - use numbers in square brackets [34]
2. Competing interests - We declare that we have no competing interests